Author's response to reviews

Title: Unusual causes of chronic abdominal pain: appendiceal mucinous neoplasm with incidental ileal endometriosis: a case report

Authors:

Takanori Kurogochi (ms03kurogochi@jikei.ac.jp)
Tetsuji Fujita (fujitetsu@jikei.ac.jp)
Naoko Iida (naoko_iida@hotmail.co.jp)
Ken Etoh (rtoken@jikei.ac.jp)
Masaichi Ogawa (masatchmo1962@yahoo.co.jp)
Katsuhiko Yanaga (kyanaga@jikei.ac.jp)

Version: 4 Date: 25 July 2012

Author's response to reviews:

Dear the Editor,

We would like to appreciate your kindness for giving us the opportunity to revise the manuscript for possible publication in the Journal of Medical Case Reports.

We revised the manuscript with the help of a native scientist to improve the quality of English. In accordance with previous publications in the Journal of Medical Case Reports, we made the contents in the abstract more concise.

The changes are indicated by underlines.

We are looking forward to hearing from you again.

With best wishes,

Takanori Kurogochi